Author's response to reviews

Title: Enterobacter nimipressuralis as a cause of Pseudobacteremia

Authors:

Dong-min Kim (drongkim@chosun.ac.kr)
Sook-Jin Jang (sibjang@chosun.ac.kr)
Ganesh Prasad Neupane (neupaneganesh@yahoo.com)
Mi-Sun Jang (sangkm3507@nate.com)
Se-Hoon Kwon (kwsh99@nate.com)
Seok-Won Kim (chosunns@hanmail.net)
Won Yong Kim (kimwy@cau.ac.kr)

Version: 7 Date: 20 October 2010

Author's response to reviews: see over
Dear Ms Roxane Rajabi
The BioMed Central Editorial Team
Email: editorial@biomedcentral.com

Re: “Enterobacter nimipressuralis as a cause of Pseudobacteremia”

Dear Editor,

Enclosed is our final proof edited manuscript “MS: 178688/1123546087” which has been corrected as per the BMC infectious diseases format. The small modifications made in our manuscript during editing time are given at the bottom of this letter.

Now we hope, our manuscript has been more interested for the clinicians and other readers of the BMC Infectious Diseases.

Thank you for your consideration.

< Corresponding author >
Seok Won Kim
Department of Neurosurgery, School of Medicine, Chosun University,
588, Seosuk-dong, Dong-gu, Gwangju-city,
501-717, Korea.
Tel : (011) 82-62- 220-3126
FAX : (011)82-62- 227-4575
E-mail : chosunns@hanmail.net

Some modifications:

In our manuscript we made small correction during proof editing time which are as follows.

1. In Abstract (page 3, line number 12), Infrequent-restriction-site PCR has been added before abbreviation IRS-PCR.
2. Table 1 has been more simplified by adding “results showing MIC (µg/ml)” in the heading and deleted E. nimipressularis from case 1, case 2 and cotton isolate.
3. MIC of “nitrofurantoin” has been corrected as per the changing of the isolates serially from 1 to 2.